Author’s response to reviews

Title: Prognostic Value of Total Tumor Volume in Patients with Nasopharyngeal Carcinoma Treated with Intensity-Modulated Radiotherapy

Authors:

Shao-Bo Liang (liangshb@sysucc.org.cn)
Jian-Jian Teng (deartjj@163.com)
Xue-Feng Hu (hxeng@fsyyy.com)
Xing-Li Yang (yangxingl@sysucc.org.cn)
Min Luo (luomin@sysucc.org.cn)
Xiao-Na Fang (fangxn@sysucc.org.cn)
Dong-Sheng Liu (ldsheng@fsyyy.com)
Yong Chen (chenyong@sysucc.org.cn)
Li-Wu Fu (fulw@sysucc.org.cn)

Version: 4 Date: 01 Jul 2017

Author’s response to reviews:

1. Please also take a moment to check our website at http://bcan.edmgr.com/ for any additional comments that were saved as attachments.

Answer:

Thank you for your helpful comment.

We have checked the website at http://bcan.edmgr.com/ for the additional comments that were saved as attachments. Also, we have revised the format of the references according to examples of the BioMed Central reference style. The amendment has been made to the manuscript (References section, line 445-562, page 21-25).
2. Please also ensure that your revised manuscript conforms to the journal style, which can be found at the Instructions for Authors on the journal homepage.

Answer:

Thank you for your helpful comment.

Based on the Instructions for Authors, we have revised the manuscript to conform to the journal style. The amendment has been made to the manuscript (Table 2, line 246-250, page 12; Table 3, line 265-270, page 13; Table 1, line 590-592, page 27-28; Figure titles and legends, line 565-588, page 26; Figure files).

3. By resubmitting your manuscript you confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.

Answer:

Thank you for this comment.

We confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.

4. Please combine the "Results" and "Discussion" section into a single "Results and discussion" section in order for the manuscript format to meet the journal guidelines.

Answer:

Thank you for this comment.

According to the submission guidelines, the "Results" and "Discussion" sections should be two separate sections. This format is also found in the recent articles published in BMC cancer. We
have contacted you to clarify if it is a must to combine the two sections but no reply is received so far. If allow, we prefer to separate the "Results" and "Discussion" into two different sections. Please feel free to let us know if you have any comment.

5. Author DSL does not appear to meet standard guidelines for authorship. We recommend that you adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. Please see http://www.biomedcentral.com/getpublished/editorial-policies#authorship for more details.

Answer:

Thank you for your helpful comment.

DSL performed the data analysis and data interpretation. The amendment has been made to the manuscript (Author Contributions section, line 415, page 19).

6. If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English.

Answer:

Thank you for your helpful comment.

We had this manuscript copyedited by a professional English editing service that specializes in scientific papers.

7. In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Cancer must have a Declarations section which includes the mandatory sub-sections listed below.
- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

Answer:

Thank you for your helpful comment.

The amendment has been made to the manuscript (Declarations section, line 424-432, page 20).